Author's response to reviews

Title: Post traumatic stress symptoms and heart rate variability in Bihar flood survivors following yoga: a randomized controlled study

Authors:

Shirley Telles (shirleytelles@gmail.com)
Nilkamal Singh (naosekpamkamal@gmail.com)
Meesha Joshi (meeshie26@gmail.com)
Acharya Balkrishna (pyp@divyayoga.com)

Version: 5 Date: 27 November 2009

Author's response to reviews:

Dear Sir or Madam,

We have attempted to revise the manuscript entitled "Post traumatic stress symptoms and heart rate variability in Bihar flood survivors following yoga: a randomized controlled study", based on the reviewers’ comments. Will be grateful to receive your response.

Thanking you,
Sincerely,
Shirley Telles, Ph.D.